Author's response to reviews

Title: Pain, power and patience - a narrative study of general practitioners' relations with chronic pain patients

Authors:

Mia Hemborg Kristiansson (Mia.Hemborg_Kristiansson@med.lu.se)
Annika Brorsson (Annika.Brorsson@med.lu.se)
Caroline Wachtler (carolinewachtler@hotmail.com)
Margareta Troein (Margareta.Troein@med.lu.se)

Version: 3 Date: 20 April 2011

Author's response to reviews:

Dear Editor,

We have made all the changes according to the suggestions below:

1. In abstract: "….students, but also in CPD Continuing Professional Development" – omit CPD!
2. "….the patient doctor relationship,. how doctors experience these" – omit dot
3. "…and what happens in consultation with them” - consultations with them..?
4. "….The results emphasize consultations skills" - consultation skills as
5. "….skills for medical students, but also in CPD (Continuing Professional Development groups)” - skills for medical students, but also in Continuing Professional Development groups (CPD)
6. .....Add to List of abbreviations: CPD

And also for your own decision: In abstract it is said: "……was to understand and illustrate what physicians experience in …"

( and also in background (last para) "The aim of this study was to understand and illustrate what physicians experience in contact with chronic pain patients and what works and does not work in these consultations.")

Can this be generalized to physicians- or should it be GPs?

Changed to GPs.

And I found a few more spelling mistakes etc which are now corrected.

We hope that our article now is fit for publishing

Best regards

Margareta Troein